# This document is downloaded from DR-NTU (https://dr.ntu.edu.sg) Nanyang Technological University, Singapore.

# Ageing and efficiency aware battery dispatch for arbitrage markets using mixed integer linear programming

Hesse, Holger C.; Kumtepeli, Volkan; Schimpe, Michael; Reniers, Jorn; Howey, David; Tripathi, Anshuman; Wang, Youyi; Jossen, Andreas

2019

Hesse, H. C., Kumtepeli, V., Schimpe, M., Reniers, J., Howey, D., Tripathi, A., . . . Jossen, A. (2019). Ageing and Efficiency Aware Battery Dispatch for Arbitrage Markets Using Mixed Integer Linear Programming. Energies, 12(6), 999-. doi:10.3390/en12060999

# https://hdl.handle.net/10356/106251

https://doi.org/10.3390/en12060999

© 2019 by the Authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).

Downloaded on 28 Aug 2022 05:13:08 SGT





# Ageing and Efficiency Aware Battery Dispatch for Arbitrage Markets Using Mixed Integer Linear Programming <sup>†</sup>

# Holger C. Hesse <sup>1,\*,‡</sup>, Volkan Kumtepeli <sup>2,‡</sup>, Michael Schimpe <sup>1</sup>, Jorn Reniers <sup>3,4</sup>, David A. Howey <sup>3</sup>, Anshuman Tripathi <sup>5</sup>, Youyi Wang <sup>6</sup> and Andreas Jossen <sup>1</sup>

- <sup>1</sup> Department of Electrical and Computer Engineering, Technical University of Munich (TUM), 80333 Munich, Germany; michael.schimpe@tum.de (M.S.); andreas.jossen@tum.de (A.J.)
- <sup>2</sup> Energy Research Institute @ NTU, Interdisciplinary Graduate Programme, Nanyang Technological University, Singapore 637371, Singapore; volkan001@e.ntu.edu.sg
- <sup>3</sup> Department of Engineering Science, University of Oxford, Oxford OX1 3PJ, UK; jorn.reniers@eng.ox.ac.uk or jorn.reniers@vito.be (J.R.); david.howey@eng.ox.ac.uk (D.A.H.)
- <sup>4</sup> Energy Technology Unit, Vlaamse Instelling voor Technologisch Onderzoek (VITO), Boeretang 200, 2400 Mol, Belgium
- <sup>5</sup> Energy Research Institute @ NTU, Nanyang Technological University, Singapore 637141, Singapore; antri@ntu.edu.sg
- <sup>6</sup> School of Electrical & Electronic Engineering, Nanyang Technological University, Singapore 639798, Singapore; eyywang@ntu.edu.sg
- \* Correspondence: holger.hesse@tum.de; Tel.: +49-89-289-26964
- † This paper is an extended version of our paper published in IEEE 18th International Conference on Environment and Electrical Engineering and 2nd Industrial and Commercial Power Systems Europe (EEEIC/I&CPS Europe), Palermo, Italy, 2018.
- ‡ These authors contributed equally to this work.

Received: 15 January 2019; Accepted: 7 March 2019; Published: 14 March 2019



Abstract: To achieve maximum profit by dispatching a battery storage system in an arbitrage operation, multiple factors must be considered. While revenue from the application is determined by the time variability of the electricity cost, the profit will be lowered by costs resulting from energy efficiency losses, as well as by battery degradation. In this paper, an optimal dispatch strategy is proposed for storage systems trading on energy arbitrage markets. The dispatch is based on a computationally-efficient implementation of a mixed-integer linear programming method, with a cost function that includes variable-energy conversion losses and a cycle-induced battery capacity fade. The parametrisation of these non-linear functions is backed by in-house laboratory tests. A detailed analysis of the proposed methods is given through case studies of different cost-inclusion scenarios, as well as battery investment-cost scenarios. An evaluation with a sample intraday market data set, collected throughout 2017 in Germany, offers a potential monthly revenue of up to 8762 EUR/MWh<sub>cap</sub> installed capacity, without accounting for the costs attributed to energy losses and battery degradation. While this is slightly above the revenue attainable in a reference application—namely, primary frequency regulation for the same sample month (7716 EUR/MWh<sub>cap</sub> installed capacity)—the situation changes if costs are considered: The optimisation reveals that losses in battery ageing and efficiency reduce the attainable profit by up to 36% for the most profitable arbitrage use case considered herein. The findings underline the significance of considering both ageing and efficiency in battery system dispatch optimisation.

**Keywords:** efficiency; storage; battery ageing; arbitrage; market; optimisation; mixed-integer-linearprogramming; piece-wise affine approximation; utility-scale; frequency regulation; primary control reserve; lithium-ion



#### 1. Introduction

An increasing number of investors are considering the installation and marketing of utility-scale battery energy storage systems (BESSs). Among others, grid ancillary services, such as the provision of a primary control reserve (PCR), are a major application of interest [1]. In this use case, revenue is typically linked to the provision of power (EUR/MW), and storage operation is closely linked to the external frequency signal. With increasing numbers of utility-scale BESSs installed and operating in control reserve markets, the attainable profits are declining, and storage system operators aim for alternative revenue streams. The operation of storage systems for energy arbitrage, where revenue is created through charging/discharging at variable electricity prices (EUR/MWh), has been discussed in various contributions; however, it has received less attention after the recent liberalisation of the European Union wholesale market [2–4]. Whilst pumped-hydro electric power plants are well-suited to monetise the diurnal fluctuations in electricity cost and have generated large revenues, the peak and off-peak patterns significantly changed as renewable energy sources (RESs) became widespread, and the price variability is often recorded on shorter timescales, rendering this application increasingly attractive for fast-reacting storage systems, such as lithium-ion batteries (LIBs) [5,6].

In the UK, the first large-scale commercial use of grid-connected batteries was to deliver an enhanced frequency response [7,8]. However, due to the limited amount of storage needed for fast frequency response and the subsequent decrease in the price of such services, attention has shifted towards multi-service portfolios and/or other applications, such as energy arbitrage, both in Europe [9,10] and in the United States [11]. In fact, a BESS is well-suited for multiple applications, in sequential or even time-parallel operation, and to improve revenues through "value stacking", making it a hotly debated topic at present [12,13]. In order to ensure the best project internal rate of return, an optimal *BESS application portfolio management strategy* may require the operation of the storage system in different applications, but should also consider the risk and costs associated with swapping of applications. However, an adequate cost analysis, including the battery dispatch, is generally not provided in the literature.

In this contribution, we will demonstrate how to optimise battery operation while considering the most relevant cost factors: battery ageing and energy losses in the system. We demonstrate the optimised storage dispatch in an exclusive arbitrage operation, but the methodology can be easily adapted for storage operation in time-parallel or sequential operation modes, as well as for other single-use applications. The economic viability of the arbitrage operation depends on the electricity price differences over time, as well as on the marginal cost of the system operation. The marginal costs can be defined as the additional costs occurring specifically due to the battery charging and discharging [14], such as energy losses and cycle-induced battery degradation during operation. In contrast, energy consumption for auxiliary components and calendar ageing effects can be considered as non-marginal, as these will be influenced only negligibly by the dispatch profile. The same holds for fixed costs (i.e. real estate, housing, air condition, wiring, and so on), which are independent of the dispatch, and are consequently excluded from this analysis. We implemented all relevant cost and revenue functions for BESS operation in an arbitrage market scenario in a mixed integer linear programming (MILP) framework, which allows us to derive the maximum profit attainable and best operational behaviour for the storage system. We, then, compared these values to the profit attainable through operation of a technically-equivalent storage system in frequency control reserve markets.

#### 1.1. Literature Review

A detailed multi-technology analysis of storage viability for electricity arbitrage was undertaken by Bradbury et al. [15]. Although their linear programming (LP) optimisation-based approach revealed unfavourable conditions for batteries in general in 2014, the work clearly underlined the significance of efficiency losses and ageing-induced costs, in the context of arbitrage profitability. A more detailed analysis of the optimal control of battery storage dispatch, with consideration of storage loss and degradation, was conducted by Koller et al. The authors considered battery degradation by an explicit piecewise affine (PWA) cost function for optimal control of a BESS [16].

Various other studies have attempted to add a more detailed degradation model to the economic analysis. Stroe et al. [17] included an empirical nonlinear degradation model in a case study for a BESS, providing primary frequency response, although no optimisation was conducted. Swierczynski et al. [18] also analysed the profitability of frequency response using a BESS. A less realistic, but simpler, degradation model was used in their work. Sun et al. [19] optimised distributed batteries for balancing the grid while including a strictly-increasing convex degradation cost, but the universal power-law type degradation model was not suitable to describe the capacity fade of a particular LIB. Sarker et al., in [20], used a MILP method, including a basic degradation model to analyse the effect of trading electricity at different power levels, but a lack of data underlying the degradation and efficiency models meant that they could not quantify the effects. Xu et al. [21] also targeted electricity trading, but using a rain-flow counting method to account for degradation. While the proposed algorithm may mimic the cycle-induced degradation of a battery to a certain extent, it did not take into account inverter and peripheral component-induced losses. Also, this work did not provide any data. Goebel et al. [22] calculated model-based dispatch strategies for a lithium-ion BESS, applied to pay-as-bid secondary reserve markets. Perez et al. [23] extended the application side by accounting for stacking applications: they analysed the effect of degradation for a battery which could be used for energy arbitrage, various balancing services, and localised peak shaving. They found that, although accounting for degradation decreases the revenue due to lower battery utilisation, the corresponding increase in lifetime more than compensated for this short-term loss. Others have used more complex battery performance and degradation models to obtain a more accurate lifetime prediction. Due to the non-linearities in these electrochemical models, a large computational effort is necessary and most studies limited their analyses to short duration (few days) use cases, instead of economic optimisation. Patsios et al. [24] analysed various control methods for a single day of usage of a battery in a peak shaving application and found that the losses could be decreased by 43% or the degradation could be halved, depending on the control method used. Lee et al. [25] studied a stand-alone photovoltaic microgrid while Weisshar et al. [26] looked at microgrids. Both reported that the electrochemical battery models increased the accuracy of the battery behaviour. Unfortunately, both studies lacked a quantitative economic analysis. Reniers et al. compared battery models of increasing complexity [27]. Their work shows that a more physically realistic modelling of battery degradation may strongly alter the battery dispatch and allow for increased BESS profitability.

Efficiency losses in containerised utility-scale storage systems have been studied in detail by Schimpe et al. [28–30], and a first demonstration, applied to the arbitrage operation, was given by the authors in a recent contribution [14]. While the work in [14] provided a framework for the techno-economic assessment of marginal cost, in this contribution we extend the existing literature by linking the cost functions of battery ageing and storage system efficiency losses with a state-of-the-art MILP framework. In this work, this framework is applied to assess the potential revenue of batteries in an arbitrage market scenario. Attainable revenue is compared to a reference case BESS application, using the same battery parameters. Although its tractability is not guaranteed, using an appropriate formulation of MILP can constitute a promising approach in solving complex optimisation problems, due to its gradually-increasing computational efficiency [31–34].

Cheng and Powell proposed an approximate dynamic programming method, considering the stochasticity and fast-changing dynamics required for frequency regulation. They could reduce the computation time to several seconds [35]. However, they assumed a constant round-trip system efficiency. In [36], a stochastic optimisation algorithm leveraging a MILP formulation for the day-ahead bidding market was proposed and compared to a traditional LP. They used binary variables to express acceptance or rejection of their bid, and it was shown that MILP was superior to LP, in terms of cumulative revenue. However, they also assumed a constant loss in charging and discharging; therefore, they did not employ any detailed loss curves for battery and power converters. It has

been shown, by Nikolakakis et al., that using such a fixed system efficiency caused overestimation of profits [37]. However, their main focus was limited to a compressed-air energy storage module for the power exchange for arbitrage. In this work, the parameters for the battery and storage models are based on a state-of-the-art lithium-iron-phosphate (LFP) commercial battery cell, specifically designed for stationary applications, and a commercially available power inverter, coupling battery systems to the low voltage alternating current (AC) grid. A discussion on the economic impact of cycle-ageing induced battery ageing, as well as the profit reduction caused by dissipation losses, is conducted and a cost-optimised operation of the BESS in an arbitrage application is provided. We also compare the revenues attainable with arbitrage to the revenue and profit attainable by operation of the storage system in a reference PCR application.

### 1.2. Structure of This Work

Section 2 gives an overview of the technical parameters for the battery storage and the inverter system. The marginal costs for the battery storage system are linked to the state-of-the-art commercial cell. The subsequent section, Section 3, provides details on the MILP optimisation procedure used to conduct the battery revenue optimisation. Section 4 then provides a detailed discussion of the dispatch optimisation algorithm, applied to test and real market data values. The concluding section, Section 5, summarises the main findings of this work and outlines directions for future research.

#### 2. Techno-Economic Framework for Battery Storage Dispatch

This section provides a framework for the profit calculations conducted in the subsequent sections. Both the application-specific revenue, as well as battery investment costs, are considered. A formulation for the marginal cost of battery operation, resulting from dissipation losses and battery ageing, is introduced, which is used to investigate the optimal dispatch for arbitrage in the subsequent section. Finally, a simulation framework to analyse battery behaviour, when delivering frequency response for PCR provision, is introduced.

#### 2.1. System Layout and Framework for Profitability Analysis

A number of factors should be considered, when the profitability of BESSs is to be compared across different applications. As motivated in a previous contribution [1], the profitability of a storage system, measured in terms of return on investment (ROI), can be estimated as:

$$ROI = \frac{J_{\text{total}}}{\mathbb{C}^{\text{inv}}} = \frac{(\mathbb{R}^{\text{APL}} - \mathbb{C}^{\text{OPEX}} - \mathbb{C}^{\text{ageing}})}{\mathbb{C}^{\text{inv}}},$$
(1)

where all cash-flow is to be taken as discounted values and  $\mathbb{R}^{APL}$  represents the net present value of the revenue attainable in a storage application,  $\mathbb{C}^{OPEX}$  is usage induced cost, and  $\mathbb{C}^{ageing}$  is a cost factor added to account for degradation.  $\mathbb{C}^{inv}$  refers to the initial investment cost of the system. A more comprehensive and quantitative discussion of cost factors and formulation of value discounting can be found in [38]. In this article, we focus on the profit in the use state of the battery only ( $J_{total}$ ) and, thus, the overall cost-revenue function can be formulated as:

$$J_{\text{total}} = \mathbb{R}^{\text{APL}}(P_{\text{AC}}) - \mathbb{C}^{\text{OPEX}}_{\text{dissipation}}(P_{\text{AC}}) - \mathbb{C}^{\text{ageing}}_{\text{usage}}(P_{\text{AC}})$$
(2)

$$\mathbb{C}_{\text{dissipation}}^{\text{OPEX}}(P_{\text{AC}}) = \mathbb{C}_{\text{dissipation,bat}}^{\text{OPEX}}(P_{\text{AC}}) + \mathbb{C}_{\text{dissipation,pe}}^{\text{OPEX}}(P_{\text{AC}}), \tag{3}$$

where  $\mathbb{C}_{dissipation}^{OPEX}$  is the superposition of battery related dissipation cost  $\mathbb{C}_{dissipation,bat}^{OPEX}$  and power electronics related dissipation cost  $\mathbb{C}_{dissipation,pe}^{OPEX}$ . As indicated in the Equations (2) and (3), all contributing factors for the cost and revenue calculations depend on the AC power dispatch of the storage system  $P_{AC}$ . A schematic of the system layout, along with the factors considered for revenue calculations, is provided in Figure 1.



**Figure 1.** Schematic drawing of a battery energy storage system (BESS), power system coupling, and grid interface components.

The flows of power to (charge, "+") and from (discharge, "-") in the battery  $(P_b^{+/-})$  are coupled to the grid power  $(P_{AC}^{+/-})$  through a bidirectional inverter. Distinct applications can be served by the system, determining the AC power flow. As the initial investment costs are not considered in this work, peripheral components of the storage system, such as the energy management system (EMS) and thermal management system (TMS), are greyed out and not considered further in this work.

#### 2.2. Battery Model and Storage System Parameters

For the storage system, we choose a commercially-available battery storage system in conjunction with an industrial-scale inverter. The Sony/muRata LIB Fortelion cell with LFP cathode has been designed particularly for stationary applications [39]. A brief summary of its main parameters is given in Table A1, and a more detailed description of the cell characteristics and its ageing behaviour can be found in [40,41]. As per [42], battery ageing can be written as a composition of two different degradation factors, cycle and calendar ageing. For marginal cost optimisation, calendar ageing is omitted.

In this study, we rely on data fits to in-house laboratory measurements for cycle ageing determination [14,40]. The model is based on single-cell measurements and scaled to mimic the behaviour of the full-sized storage system. We anticipate a constant battery cell temperature of 25 °C and neglect the state of charge (SOC) dependence of the open circuit voltage. These simplifications may be justified for LFP cells, due to the minimal voltage variation of the cell within the operation limits of 10–95% SOC and a limitation of battery charge/discharge power (C-rates  $\leq$  1), as used in this study and explained further in the following section. At room temperature and with battery currents within cell specifications, the cycle ageing is mostly determined by the energy throughput and is independent of battery SOC. In the following, we use a formulation of full equivalent cycle (FEC) based on an incremental change of SOC:

$$FEC(t) = 0.5 \times \int_0^t \left| \frac{d}{d\tau} SOC(\tau) \right| d\tau \approx 0.5 \times \int_0^t \frac{|P_{\rm b}(\tau)|}{E_{\rm nom}} d\tau \approx 0.5 \times \int_0^t C_{\rm r}(\tau) d\tau, \tag{4}$$

where  $P_b$  denotes the power charged to or discharged and  $E_{nom}$  the nominal energy of the battery. As such, one unit of FEC denotes one full charge and discharge of the battery. A change in FEC may be calculated using the power-to-energy ratio (PER) or, alternatively, by a "pseudo C-rate"  $C_r$ , as the battery voltage is assumed constant.

Despite investment costs for the battery system not being considered directly in the optimisation, these costs will influence the degradation costs, as apparent in Equation (5).

$$\mathbb{C}_{\text{usage}}^{\text{ageing}}(P_{\text{AC}}) = \frac{-\Delta SOH}{(1 - k_{\text{EOL}})} \cdot \mathbb{C}_{\text{bat}}^{\text{inv}}.$$
(5)

In this equation,  $\Delta SOH$  represents the normalised state of health (SOH) losses. We assume a battery end of life (EOL) of 80%, which corresponds to  $k_{\text{EOL}} = 0.8$ , in accordance with the literature [14]. For the base case, a battery retail cost of  $\mathbb{P}_{\text{bat}}^{\text{base}} = 500 \text{ EUR/kWh}_{\text{cap}}$  was chosen, being approximately 20% below recent single-unit pricing information (excluding VAT) [43]. Nevertheless, as battery prices are declining rapidly, a potential future scenario with halved cost ( $\mathbb{P}_{\text{bat}}^{\text{future}} = 250 \text{ EUR/kWh}_{\text{cap}}$ ) is also considered [44].

A bidirectional Siemens *Sinamics S120* inverter with rated power of 36 kW, which connects to the low voltage 400 VAC grid, was used to link the storage unit to the main grid. A full parametrisation of the inverter, based on measurements taken at DC voltages of 600–750 VDC and at 0% to 100% of the nominal power for both charge and discharge, was conducted in [14]. We use the corresponding look-up tables as a function of voltage, relative power, and power flow direction for the subsequent calculations.

#### 2.3. Market Data and Battery Cost Analysis

Publicly-available energy market data, recorded in Germany, was used to conduct the optimisation. Energy prices were obtained at the European Energy Exchange through market clearing of aggregated supply and demand curves. While intraday market (IDM) prices are set on a quarter-hour basis continuously throughout the day and subject to strong variations, day ahead market (DAM) prices are set on an hourly basis and are attained by previous day auctioning. The course of a typical pricing signal is visualised in Figure 2, and a more detailed version is also given in the Appendix A, in Figure A1.



**Figure 2.** Price distribution for (**a**) the intraday market and (**b**) the day-ahead market [45]. Red lines indicate median values, blue boxes lower and upper quartiles, black whiskers indicate one standard deviation, and red marks indicate outliers.

For the IDM, an average cost of 34.68 EUR/MWh was recorded for the year 2017, used also as an input parameter for the estimation of PCR operational costs.

#### 2.4. Reference Scenario Operation

In order to relate profits attainable for BESS operation in arbitrage markets to an alternative mode of operation, we simulated the operation of a technically-equivalent storage system (with the same battery and inverter technology but an adapted PER) in the PCR operation mode. In contrast to the aforementioned arbitrage operation, PCR prices are archived on a weekly basis through pay-as-bid auctioning. This binds a storage system (as a winning tender) to operation following the grid frequency signal, with few degrees of freedom [46]. The battery storage evaluation tool *SimSES* (Open Source version available for download at www.simSES.org) is used for this purpose [1]. The tool allows for time-series modelling of battery storage systems in different application scenarios, with detailed modelling of battery ageing, as well as dissipation losses, occurring in the battery and the inverter.

In order to meet European regulatory constraints and to allow for continuous PCR operation it is mandatory to set appropriate PER and SOC limits. A PER of 0.8, is applied (i.e., backing up a power provision of 1 MW with 1.25 MWh storage size). Furthermore, the SOC limitations required 30 min of continuous maximum power provision, and an operation strategy incorporating all degrees of freedom for SOC stabilisation was used in line with the previous work [46]. Using these parameters, SimSES provides a time-series evaluation (as in Figure A4) of the storage system operation, along with an economic evaluation. Table 1 provides a short summary of average prices attained for a weekly-tendered PCR power provision for the years 2015–2017, based on publicly available market data [47,48].

**Table 1.** Primary control reserve (PCR) market data summary. Tender prices displayed are mean values for winning bids of weekly auctions. Extremum weeks and average yearly values are displayed.

| Parameter                          | Weekly Price (EUR/MW) |
|------------------------------------|-----------------------|
| Yearly average tender price (2015) | 3724                  |
| Yearly average tender price (2016) | 2494                  |
| Yearly average tender price (2017) | 2489                  |
| Week min. tender price (2017)      | 2094                  |
| Week max. tender price (2017)      | 2510                  |

As directly apparent from the simple data analysis conducted, the profitability of this use case declines and a battery operation in alternative applications (e.g., arbitrage marketing) could be analysed by a storage system owner.

#### 2.5. Marginal Cost for Battery Operation

Based on the results of the previous work, we consider dissipation losses and ageing losses of the aforementioned storage system [14]. Dissipation losses may occur from the battery cells, as well as from the AC power converter system. Based on laboratory experiments, a current-dependent ageing formulation was derived, which describes the capacity fade of the battery cells. Figures 3a,b depict the dissipation and ageing losses for the system with respect to the AC power. A conversion to the C-rate for charging or discharging was also calculated, taking the losses of the power electronics into account. The energy conversion losses in the battery were calculated using an equivalent circuit model, featuring an open circuit voltage and a single resistance using Ohm's law, as is given in Equation (6).

$$P_{\rm loss}^{\rm b}(P_{\rm AC}) = R_{\rm CH/DCH} \cdot I^{\rm b}(P_{\rm AC})^2.$$
<sup>(6)</sup>

The parameters were implemented for the aforementioned commercial battery cell, with values as provided in [14,40] (muRata/Sony Fortelion LFP cell,  $R_{CH} = 46.66 \text{ m}\Omega$ ,  $R_{DCH} = 50.29 \text{ m}\Omega$ ,  $U_{OCV} =$ 



**Figure 3.** Loss factor (**a**) and battery ageing (**b**) analysis with dependence on alternating current (AC)-side system power. The nominal power rating of 36 kW for the inverter is indicated with a dotted line. The x-axis shows both the AC power and the C-rate in charge/discharge directions, separately.

The energy conversion losses in the power electronics are based on the power-loss measurements of the previously described commercial inverter and look-up table values are taken as input for this study [14]. Round-trip efficiency values, calculated from the data, are given in Figure 4.



Figure 4. Round-trip efficiency values for battery, power electronics, and overall BESS.

### 3. Optimisation Framework

MILP is a powerful tool that can be used to approximate and optimise complicated problems involving non-linear cost functions and constraints. Although the solution time is not guaranteed to be ideal, with recent advancements in solvers, the solution can be found in a reasonable amount of time for many problems [31,32,49–51]. In our work, three components (namely the dissipation loss in the battery, the dissipation loss of the inverter, and cycle ageing for charging) are formulated through a PWA approximation. Modelling the battery discharge loss is straightforward, due to its linear dependence on  $P_{AC}$ .

Energies 2019, 12, 999

In order to maximise profit from buying and selling energy on the arbitrage markets, the objective function, introduced in Equation (2), may be reformulated as a minimisation problem, given by Equation (7),

$$-J_{\text{total}} = \underbrace{-\mathbb{R}_{\text{arbitrage}} + \mathbb{C}_{\text{dissipation}}}_{c_{\text{en}} \cdot P_{\text{AC}} \Delta t} + \underbrace{\mathbb{C}_{\text{ageing}}}_{c_{\text{ag}} \cdot (-\Delta SOH)}$$
$$= \sum_{k=1}^{N_{h}} \left( c_{\text{en}}(k) \cdot P_{\text{AC}}(k) \Delta t + c_{\text{ag}} \cdot (-\Delta SOH(k)) \right), \tag{7}$$

where  $P_{AC}$  and  $J_{total}$  denote net power exchange with the grid and net profit function, assuming an a priori known arbitrage price signal  $c_{en}$ , represented by the cost of energy [EUR/kWh] over a certain prediction horizon,  $N_h$ . The ageing-related cost  $c_{ag}$  is given as cost of unit ageing [EUR/ $\Delta$ SOH].  $\Delta$ SOH and  $\Delta t$  show the battery degradation in percent and time step in hours, respectively. As given in (7), it is worth mentioning here that the dissipation losses  $\mathbb{C}_{dissipation}$  are embedded inside the formulation, given by the subsequent system dynamics. The parameters used for the optimisation can be found in Table A2.

#### 3.1. Battery Dynamics Modelling

In this subsection, battery model equations, applied  $\forall k \in \mathbb{I}_{[1:N_h]}$ , are presented. First, the model is defined by disambiguation of the battery charging power  $P_b^+$  and discharging power  $P_b^-$ , which are limited by the maximum battery power  $P_{\text{batmax}}$ .

$$0 \le P_{\rm b}^+(k) \le s_{\rm b}(k) \cdot P_{\rm batmax} \tag{8}$$

$$0 \le P_{\mathsf{b}}^{-}(k) \le (1 - s_{\mathsf{b}}(k)) \cdot P_{\mathsf{batmax}},\tag{9}$$

where  $s_b(k) \in \mathbb{Z}_2$  is a binary variable indicating the status of the battery. The battery power is linked with the AC-side power,  $P_{AC}$ , by Equations (10)–(13).

$$P_{\rm AC}(k) = P_{\rm AC}^+(k) - P_{\rm AC}^-(k), \tag{10}$$

$$P_{\rm AC}^+(k) = P_{\rm b}^+(k) + L^+(k), \tag{11}$$

$$P_{\rm AC}^{-}(k) = P_{\rm b}^{-}(k) - L^{-}(k), \tag{12}$$

$$0 \le P_{\rm AC}^+(k), \quad P_{\rm AC}^-(k) \le P_{\rm invmax},\tag{13}$$

where  $P_{AC}^+$ , and  $L^+$  are the charging power and charging losses, respectively, and  $P_{AC}^-$ ,  $L^-$  denotes the discharging power and discharging losses on the AC side. The AC side power  $P_{AC}^{+/-}$  is constrained by the maximum power of the inverter  $P_{invmax}$ , for which detailed information about loss modelling is given in Appendix A.2. Lastly, the auxiliary variables net battery power  $P_{net}$  and absolute battery power  $P_{gross}$  are defined by (14) and (15), respectively.

$$P_{\rm net}(k) = P_{\rm b}^+(k) - P_{\rm b}^-(k), \tag{14}$$

$$P_{\text{gross}}(k) = P_{\text{b}}^{+}(k) + P_{\text{b}}^{-}(k).$$
 (15)

The main part of the battery dynamic model for the energy stored in the battery is given in Equations (16)-(19) [52].

$$E_{\text{bat}}(0) = 0 \tag{16}$$

$$E_{\text{bat}}(k) = E_{\text{bat}}(k-1) + P_{\text{net}}\Delta t$$
(17)

$$0 \le E_{\text{bat}}(k) \le E_{\max}(k) \tag{18}$$

$$E_{\max}(k) = SOH(k) \cdot E_{\text{nom}} \cdot (SOC_{\max} - SOC_{\min}).$$
<sup>(19)</sup>

Here,  $E_{\text{bat}} \in \mathbb{R}^{(N_h+1)}$  denotes the usable energy stored in the battery, which is restrained by a diminishing upper limit  $E_{\text{max}}$ . Although, in many studies, stored energy is defined by SOC, as in [52], in this study we have leveraged the representation by using  $E_{\text{bat}}$  to capture capacity loss effects due to ageing, as in [53]. Therefore, we defined an intermediate variable for the maximum energy that can be stored using the SOC limits  $SOC_{\text{max}}$  and  $SOC_{\text{min}}$ , as well as the nominal energy of the battery  $E_{\text{nom}}$ . The variable is then decreased, in proportion with the state of health *SOH*, and updated by Equations (20)–(24).

$$SOH(0) = 1 \tag{20}$$

$$FEC(0) = 0 \tag{21}$$

$$SOH(k) = SOH(k-1) + \Delta SOH(k)$$
(22)

$$FEC(k) = FEC(k-1) + \Delta FEC(k)$$
(23)

$$\Delta FEC(k) = C_{\rm r}(k)\Delta t/2. \tag{24}$$

As charging and discharging powers are recorded separately,  $C_r$  is defined by the following equations; where  $C_r^+$  and  $C_r^-$  denote the charging and discharging C-rates, respectively, and  $C_{rmax}^+$  and  $C_{rmax}^-$  are the upper limits on the charging and discharging C-rates, respectively.

$$C_{\rm r}(k) = C_{\rm r}^+(k) + C_{\rm r}^-(k)$$
(25)

$$C_{\rm r}^+(k) = P_{\rm b}^+(k) / E_{\rm nom}$$
 (26)

$$C_{\rm r}^{-}(k) = P_{\rm b}^{-}(k) / E_{\rm nom}$$
 (27)

$$0 \le C_r^+(k) \le C_{\text{rmax}}^+ \tag{28}$$

$$0 \le C_{\rm r}^-(k) \le C_{\rm rmax}^-.\tag{29}$$

#### 3.2. Battery Ageing Modelling

The cycle ageing of the cell used for this study can be modelled by a semi-empirical model fitted to experimental data, which was first presented in [40] and applied to marginal cost calculations in [14]. The convex fit on charging (as in Figure 3b) is approximated using PWA in Equation (31), as detailed in Appendix A.1. For discharging, a power-independent linear function of  $\Delta FEC$  is obtained, as in Equation (32). The capacity fade is attributed mostly to battery internal loss mechanisms, such as lithium plating and growth of a solid electrolyte interphase layer. A more detailed discussion of these physical processes can be found in [40]. The work in [40] also provides experimentally-confirmed values for the constant coefficient  $a_{SOH}$ , as well as the function  $f_{ag}$ .

$$\Delta SOH(k) = \Delta SOH^{+}(k) + \Delta SOH^{-}(k)$$
(30)

$$\Delta SOH^{+}(k) = f_{ag}(P_{AC}^{+}(k))\Delta t \tag{31}$$

$$\Delta SOH^{-}(k) = a_{\rm SOH} \cdot \Delta FEC(k). \tag{32}$$

### 4. Results and Discussion

To explore the range of results using the optimisation approach, we differentiate between three different cases: Case 1, where all losses are neglected; Case 2, where battery energy efficiency as well as battery cycle ageing are considered; and, lastly, Case 3, where the dissipation losses of the power-electronic equipment are added into case 2.

After developing our optimisation framework using YALMIP [54] and Gurobi [55], as given in Appendix A.1, its validity was verified against a test signal with a predictable outcome (see Figure 5). The test signal (Figure 5a) begins at an initial value of 0.5 EUR/kWh and gradually diverges. While, in the ideal state (Case 1), the storage system follows price fluctuations directly; we observe a battery dispatch only after a certain threshold value when additionally battery losses and ageing (Case 2), as

well as power electronics losses (Case 3), are considered. In line with the cost function formulations, the power dispatch increases, step by step, as an increasing cost burden of higher power dispatch is overcome at strong price variations. The optimiser begins dispatching the battery for energy trade only when it can cover the costs of losses with the expected profit. In fact, as is apparent from Figure 5b, in the case considering all costs (Case 3), the system starts to dispatch much later than in the case where inverter losses are neglected. At larger price differences ( $\Delta p \ge 0.1$  EUR/kWh) between subsequent time steps, all cases converge to the same solution, namely charging and discharging at maximum power, limited by the C-rate and the SOC limits. As all of our cost function formulations are independent from battery SOC, the dispatch leads to an undulation around arbitrary SOC points with an increasing amplitude (see Figure 5c).



**Figure 5.** Results of optimising dispatch for the three case study scenarios (Case 1: "All losses neglected"; Case 2: "Battery ageing and battery dissipation losses considered"; and Case 3: "Battery ageing, dissipation losses of both power electronics and battery considered". Subfigures indicate (a) Price signal; (b) AC power dispatch; and (c) state of charge.

After validating the algorithm, real-world price data sets were fed to the optimiser. The method was applied to one month of real-world data obtained from the German EPEX intraday (IDM) and day-ahead (DAM) auction markets [45]. Numerical results from the data sets are compiled in Table 2, along with a comparison to an alternative revenue source; namely PCR using the same set of technical parameters for the storage system. As such, the results can be used to give a direct comparison of the profits attainable. The profit values, in Table 2, are normalised per month and one MWh<sub>cap</sub> of battery capacity.

| Application and Dataset  | Cash Flow under Different Cases [EUR/MWh <sub>cap</sub> /month] |         |         |         |         |  |  |
|--|---|---------|---------|---------|---------|--|--|
| Application and Dataset  | Case 1  | Case 2  |         | Case 3  |         |  |  |
| Battery Cost (per kWh <sub>cap</sub> )   |   | 250 EUR | 500 EUR | 250 EUR | 500 EUR |  |  |
| Intraday Market  |   |         |         |         |         |  |  |
| Revenue $(\mathbb{R}^{APL})$   | 8762  | 8269    | 7908    | 8068    | 7671    |  |  |
| PE efficiency losses $\left(\mathbb{C}_{\text{dissipation,pe}}^{\text{OPEX}}\right)$ | 0   | 0       | 0       | 572     | 525     |  |  |
| Battery efficiency losses $\left(\mathbb{C}_{dissipation, bat}^{OPEX}\right)$        | 0   | 914     | 812     | 802     | 707     |  |  |
| Cycle ageing losses $\left(\mathbb{C}^{	ext{ageing}}_{	ext{usage}} ight)$            | 0   | 646     | 931     | 610     | 864     |  |  |
| Net profit $(J_{total})$   | 8762  | 6709    | 6165    | 6085    | 5575    |  |  |
| Day Ahead Market   |   |         |         |         |         |  |  |
| Revenue $(\mathbb{R}^{APL})$   | 1724  | 1618    | 1599    | 1560    | 1528    |  |  |
| PE efficiency losses $\left(\mathbb{C}_{\text{dissipation,pe}}^{\text{OPEX}}\right)$ | 0   | 0       | 0       | 148     | 141     |  |  |
| Battery efficiency losses $\left(\mathbb{C}_{dissipation, bat}^{OPEX}\right)$        | 0   | 143     | 135     | 132     | 119     |  |  |
| Cycle ageing losses $\left(\mathbb{C}^{	ext{ageing}}_{	ext{usage}} ight)$            | 0   | 64      | 114     | 53      | 91      |  |  |
| Net profit $(J_{total})$   | 1724  | 1411    | 1351    | 1227    | 1177    |  |  |
| PCR (Reference Operation)  |   |         |         |         |         |  |  |
| <i>Revenue</i> $(\mathbb{R}^{APL})$  | 7716  | 77      | '16     | 7       | 716     |  |  |
| PE efficiency losses $\left(\mathbb{C}_{\text{dissipation,pe}}^{\text{OPEX}}\right)$ | 0   | (       | 0       | 3       | 366     |  |  |
| Battery efficiency losses $\left(\mathbb{C}_{dissipation, bat}^{OPEX}\right)$        | 0   | 2       | 4       |         | 24      |  |  |
| Cycle ageing losses $\left(\mathbb{C}^{	ext{ageing}}_{	ext{usage}} ight)$            | 0   | 18      | 36      | 18      | 36      |  |  |
| Net profit $(J_{total})$   | 7716  | 7674    | 7656    | 7308    | 7290    |  |  |

Table 2. Cash flow for storage operation in three use cases based on real-world data (2017).

As apparent from Table 2, the arbitrage cost, revenue, and profits were widely varying, depending on the consideration of ageing and dissipation losses. Similarly, not only the costs but also the profit and, ultimately, revenue were affected, when switching between the battery cost assumptions (250 EUR/kWh or 500 EUR/kWh) for otherwise-unchanged parameter settings. In fact, the algorithm adapts its dispatch strategy to optimise the marginal cost. It is also evident that DAM profits were significantly lower, compared to IDM profits. This is attributed both to overall lower price differences in the day ahead market, as well as a lower sampling rate (15 min versus 1 h), resulting in less battery dispatch operations within the sample time. The table also reveals differences between both arbitrage markets analysed. While ageing is one of the largest contributors to cost in all the IDM operation cases under investigation, for DAM applications, the dissipation losses of both the battery and inverter infer a higher cost burden. This is attributed to an overall lower relative power during IDM operation, the typically low power electronics efficiency, and high relative losses under partial load operation. Overall efficiency and energy throughput values are also given in Table A3.

To better visualise the power distribution of the various scenarios, refer to Figure 6, as well as Figures A2 and A3. In Figure 6, the plots show a cumulative frequency distribution versus battery C-rate for the 500 EUR/kWh battery cost scenario. One can clearly see that the optimiser tends to favour maximum C-rates in the Case 1 scenario (green lines), where ageing and losses are not

considered. If ageing and battery dissipation are taken into account (Case 2, red line) high C-rates are avoided, in order to keep the battery-induced ageing and dissipation loss costs at an acceptable level (as in Figure A5). This comes at the cost of a slightly reduced overall trading revenue. Looking back to Table 2, this results in a reduction from 8762 EUR/MWh<sub>cap</sub> down to 7908 EUR/MWh<sub>cap</sub> monthly revenue for IDM arbitrage between Case 1 and Case 2 (500 EUR/kWh<sub>cap</sub>). When subtracting the costs of battery dissipation losses and degradation, this results in a monthly net profit of 6165 EUR/MWh<sub>cap</sub>. For Case 3 (blue lines), where power electronics losses (being particularly prominent at partial load conditions) are additionally taken into account, the optimiser tries to find a compromise and avoids both high C-rates and low C-rates. As a result, compared to Case 2 (red lines), in Case 3 (blue lines) the dwell time at no load conditions is significantly higher. In the IDM application, the power electronics losses comprise of an additional 5.99% loss, which is 25.04% of the total cost. In DAM, the effects are even more pronounced: power electronics losses account for an 8.15% reduction of efficiency and comprise 40.01% of the total losses. We, therefore, conclude that, for battery arbitrage market studies, dissipation losses should always be considered for a net profit calculation.



**Figure 6.** Cumulative probability distributions of C-rate for: (**a**) The intraday market (IDM); (**b**) the day-ahead market (DAM); and (**c**) the PCR.

For the reference PCR operation (Figure 6c), where the operation closely follows the frequency signal, we observe an overall lower impact of ageing and battery efficiency losses, which will increase with higher battery C-rates (see Figure 3). For the Case 3 scenario, however, a significant cost burden (366 EUR/MWh<sub>cap</sub>) is associated with the power electronics losses, being prominent particularly at partial load conditions. From a system engineering viewpoint, these findings highlight the importance of lowering the dissipation losses of the power electronics and pushing for a higher cycle life of batteries, when large scale storage systems are to be used in an arbitrage operation. At the same time, the cost and revenue analysis also highlights the importance and necessity of taking ageing and efficiency losses into account when designing a profit-optimal battery dispatch strategy.

Economic results normalised to the maximum revenue are shown in Figure 7, for each scenario. When looking at the IDM and DAM bar plots, we see at first glance that, for both applications, not modelling the power conversion losses and ageing costs would cause a significant overestimation of the profits (Case 1 versus Case 3). Furthermore, the normalised bar plots illustrate which cost contributor should be tackled first, in order to achieve improved net profits for the case of considering all cost contributors (Case 3): while power electronic losses outweigh other costs for the PCR application, ageing losses are the largest cost contributor for the DAM application; at a battery cost of 500 EUR/kWh<sub>cap</sub>. Ageing becomes a less-important cost contributor for the less energy-throughput demanding DAM application. A potential decrease of battery investment cost will further reduce the impact of ageing costs to the net profit.



**Figure 7.** Comparison of revenue and losses resulting from battery ageing, as well as dissipation losses, for the storage system.

#### 5. Conclusions and Outlook

This paper provides an optimisation framework for arbitrage battery dispatch, along with a full parametrisation relevant to a commercially available battery storage system including a LFP-type lithium-ion battery and an industry-scale inverter component. Real-world data for the arbitrage markets and the provision of PCR was used as an input, to derive the revenue and costs of the storage system operation under various conditions and constraints. To systematically describe the ageing and efficiency, a previously-derived set of non-linear functions, based on in-house experimental data, was incorporated into the system with the help of PWA approximations and an open-source toolkit. The discontinuities and nonconvexities in this PWA approximation were expressed using an MILP formulation, where a specialised algorithm was employed to reduce the number of integer and redundant real variables.

A test data set is used to validate the applicability of the algorithm. Depending on the operational costs considered (Cases 1–3), the operation of the storage system adapts and finds an ideal compromise of cost and revenue generation by arbitrage dispatch. Thereafter, the method is applied to real-world data sets and net profits are obtained for various scenarios. The quarter-hourly IDM with significant price fluctuations is more profitable for battery arbitrage, compared to a DAM operation with an hourly pricing scheme. The profit attainable for storage operation in an IDM arbitrage operation is found to be on a similar level and, in some scenarios analysed, even slightly above the revenue made in the reference PCR operation.

In order to consider the rapid decrease in the battery costs currently observed, two price scenarios are analysed. While. in the base case (500 EUR/kWh<sub>cap</sub>), the costs associated with battery ageing reduce attainable arbitrage revenue (intraday market, Case 3) by 11.2%, their impact is slightly less pronounced in an optimistic future scenario with a halved battery cost (revenue reduction by 7.6%). In all cases considered, the sum of costs due to dissipation losses even outweighs the cost for battery cycle ageing. However, it is worth mentioning that the cycle life of the battery cell used for this study is far above average, and the results could change for a different battery cell.

The results obtained with the real-world data clearly highlight the importance of modelling efficiency losses and ageing, along with revenue maximisation, in arbitrage markets. For the data sets used herein and when taking all costs into account, an arbitrage operation is still slightly less profitable when using the storage in the reference PCR application. However, this situation might change in the near future: PCR prices might continue to fall, whereas an increasing share in variable renewable generation might induce stronger fluctuations on the wholesale electricity markets.

While it provides a versatile framework for a direct comparison of a BESS in competing applications, there are some limitations to this study. The market data is assumed to be known a priori for the entire optimisation horizon. However, in reality, forecasts will not match market results

perfectly. Therefore, our results reflect only the maximum attainable profit. The accuracy of the profit estimation can be further analysed, regarding forecasting and adaptive algorithms. Only marginal costs are considered in this work. While this allows one to find the most profitable applications for a storage system, it can not provide a full profitability analysis. The fixed costs of the storage system, as well as calendar ageing, should be included to derive the return on investment and the profitability of the system over the battery lifetime. An improved battery performance model could be used, which considers the SOC and current dependence of the battery terminal voltage, leading to increased model accuracy. More advanced ageing formulations could be implemented, such as taking battery temperature and SOC dependence into account. Sophisticated models, based on physico-chemical reactions, may need to be used to capture more complex and non-linear ageing phenomena [56,57].

Although MILP-based optimisation has a drastically increased speed, compared to a simplistic brute-force approach, the formulations are still memory-consuming (one month simulation of IDM in Case 3 takes approximately 5–10 h on a Windows 10 computer with an i7-6700HQ CPU at 2.60 GHz and 16 GB 2333 MHz RAM using Gurobi 8.1 with a mixed integer programming gap of 0.05%), and we were unable to conduct simulations over a time horizon of one year. There are a couple of different methods that can be used to enhance the scalability of the simulations. Use of receding horizon (or rolling horizon)-based methods, namely model predictive control (MPC), could be a method of choice for future investigations. The MPC procedure reduces the computation time by using a small horizon and creating an artificial closed loop system [58,59]. Since the price forecast may be imprecise and/or unavailable for long time frames, one may also prefer MPC. Utilising an integrated closed loop controller might enable the system to better handle uncertainties in price signals [60]. Another approach would be using a better decomposition method to create an economic MILP formulation. As the problem can be formulated as a difference of convex PWA functions [61], it can potentially be solved more efficiently by specific methods, as discussed in [62].

Author Contributions: Conceptualization, H.C.H.; Methodology, V.K.; Project administration, H.C.H.; Software, V.K.; Supervision, D.A.H., A.T., Y.W., and A.J; Validation, H.C.H.; Writing—original draft, H.C.H., V.K., M.S., and J.R.; Writing—review & editing, H.C.H., V.K., M.S., D.A.H., and J.R.

Acknowledgments: This work was supported by by TUM (Technical University of Munich, Germany) and ERI@N (Energy Research Institute at Nanyang Technological University, Singapore) within the joint project agreement ICER (International Center of Energy Research); the international travel funding of DIMG-EMS; VITO (Vlaamse Instelling Voor Technologisch Onderzoek) and EnergyVille, Belgium. This work was also supported by the German Research Foundation (DFG) and the Technical University of Munich (TUM) in the framework of the Open Access Publishing Program. The responsibility for this publication rests with the authors.

Conflicts of Interest: The authors declare no conflict of interest.

## Abbreviations

The following abbreviations are used in this manuscript:

- AC alternating current
- BESS battery energy storage system
- DAM day ahead market
- EMS energy management system
- EOL end of life
- FEC full equivalent cycle
- IDM intraday market
- LFP lithium-iron-phosphate
- LIB lithium-ion battery
- LP linear programming
- MILP mixed integer linear programming
- MPC model predictive control
- PCR primary control reserve

- PER power-to-energy ratio
- PWA piecewise affine
- RES renewable energy source
- ROI return on investment
- SOC state of charge
- SOH state of health
- TMS thermal management system

# Appendix A

Table A1. Technical parameters of the battery and inverter system used for this study.

| Parameter             | Value                               |
|-----------------------|-------------------------------------|
| Battery type          | muRata Fortelion (Sony US26650FTC1) |
| Cathode chemistry     | Lithium-iron-phosphate              |
| Nominal cell voltage  | 3.2 V                               |
| Nominal cell capacity | 3 Ah                                |
| Storage capacity      | 36 kWh                              |
| Operation limits      | [10 · · · 95] % SOC                 |
| Âgeing model          | Empirical cycle ageing model        |
| Inverter type         | Siemens Sinamics s120               |
| Inverter power rating | 36 kW (nominal) 43.2 kW (maximum)   |

Table A2. Optimisation parameters.

| Parameter              | Value   | Unit             |
|------------------------|---|------------------|
| c <sub>bat</sub>       | 500 or 250  | EUR/kWh          |
| Cag                    | $c_{bat} \cdot E_{nom} / (1 - k_{EOL}) = 9 \cdot 10^4 \text{ (for } c_{bat} = 500) \text{ or } 4.5 \cdot 10^4 \text{ (for } c_{bat} = 250)$ | $EUR/\Delta SOH$ |
| $k_{EOL}$              | 0.8   | N.A.             |
| $E_{nom}$              | 36  | kWh              |
| $P_{batmax}$           | 36  | kW               |
| Pinvmax                | 43.2  | kW               |
| $SOC_{max}$            | 0.95  | N.A.             |
| $SOC_{min}$            | 0.1   | N.A.             |
| $N_h$                  | 30 · 96 (IDM) or 30 · 24 (DAM)  | N.A.             |
| $\Delta t$             | 0.25 (IDM) or 1 hour (DAM)  | h                |
| $C^+_{rmax}, C^{rmax}$ | 1   | N.A.             |
| a <sub>SOH</sub>       | $-3.18 \cdot 10^{-7}$   | $\Delta SOH/FEC$ |



Figure A1. Price distribution for (a) the intraday market; and (b) the day-ahead market.



**Figure A2.** Histogram plots of C-rate for (**a**) the intraday market; (**b**) the day-ahead market; and (**c**) the primary control reserve.



**Figure A3.** For PCR, DAM, and IDM in the Case 3 scenario (considering all losses and ageing): (a) Histogram plots; and (b) cumulative distribution.

| Application and Dataset                | System Indicators under Different Cases |         |         |         |         |  |  |
|--|---|---------|---------|---------|---------|--|--|
| Application and Dataset                | Case 1                                  | Cas     | se 2    | Case 3  |         |  |  |
| Battery Cost (per kWh <sub>cap</sub> ) |   | 250 EUR | 500 EUR | 250 EUR | 500 EUR |  |  |
| Intraday Market                        |   |         |         |         |         |  |  |
| Energy throughput [kWh]                | 21,647                                  | 16,073  | 14,498  | 14,461  | 12,924  |  |  |
| Full equivalent cycle [-]              | 301                                     | 223     | 201     | 201     | 180     |  |  |
| Mean round-trip efficiency [%]         | 100                                     | 93.3    | 93.6    | 88.7    | 89.0    |  |  |
| Day Ahead Market                       |   |         |         |         |         |  |  |
| Energy throughput [kWh]                | 5814                                    | 3656    | 3418    | 2721    | 2487    |  |  |
| Full equivalent cycle [-]              | 81                                      | 51      | 47      | 38      | 35      |  |  |
| Mean round-trip efficiency [%]         | 100                                     | 96.6    | 96.7    | 90.6    | 90.6    |  |  |
| PCR (Reference Operation)              |   |         |         |         |         |  |  |
| Energy throughput [kWh]                | -                                       | _       | -       | 2168    |         |  |  |
| Full equivalent cycle [-]              | -                                       | -       | -       | 2       | 8       |  |  |
| Mean round-trip efficiency [%]         | -                                       | -       | -       | 80      | ).7     |  |  |

|--|



**Figure A4.** SimSES modelling of a 30 day PCR operation. The battery model is set to the lithium-iron-phosphate (LFP) cathode chemistry, and ageing is limited to cycle ageing solely.



Figure A5. Ageing for: (a) Test data; (b) the intraday market; and (c) the day-ahead market.

The power of MILP lies in its ability to model highly complex functions using piecewise affine (PWA) approximations. Therefore, the nonlinear parts in the model, the ageing and loss functions, are modelled by suitable PWAs. As shown in Figure A6,  $f_{ag}$  is a nearly-quadratic (but not exactly) function of  $P_{AC}^+$ . Therefore, these nonlinearities must be approximated by a series of PWA functions to use the MILP approach. However, since it is a convex function, the formulation can be done without using integer variables, which reduces the computational burden substantially.



Figure A6. Piecewise affine (PWA) curve fitting for the semi-empirical cycle ageing curve (charging).

A sample PWA function is given in the form of Equation (A1) and the actual function can be approximated by selecting the maximum value of the pieces, as in Equation (A2) [63].

$$\tilde{f}(z) = \begin{cases} a_{1}z + b_{1} & \text{if } r_{0} \leq z \leq r_{1} \\ \vdots \\ a_{i}z + b_{i} & \text{if } r_{i} \leq z \leq r_{i+1} \\ \vdots \\ a_{N}z + b_{N} & \text{if } r_{N-1} \leq z \leq r_{N} \end{cases}$$
(A1)  
$$f \approx \max_{a_{i},b_{i}} \left(\tilde{f}(z)\right).$$
(A2)

This formulation can be converted into epigraph form, as in Equation (A3) [63]. In the epigraph form, a variable called  $f_{PWA}$  is defined as an upper boundary for all PWA functions. When an expression involving  $f_{PWA}$  is minimised,  $f_{PWA}$  will be minimised until it intercepts the maximum value amongst the PWA functions. Hence, it can accurately represent a convex PWA function.

$$\min_{z} \left( \tilde{f}(z) \right) \equiv \begin{bmatrix} \min_{z} \left( f_{\text{PWA}} \right) & \text{s.t.} & a_{i}z + b_{i} \leq f_{\text{PWA}} & \forall i \in \mathbb{I}_{[1:N]} \end{bmatrix}.$$
(A3)

After obtaining an appropriate form for approximation, the coefficients  $a_i$  and  $b_i$ , and the breakpoints  $r_i$ , in Equation (A1), can be found by solving a nonlinear programming problem, defined by Equation (A4) for a pre-defined *N* number of pieces [52]. Therefore, an open-source toolbox, developed by Alexander Szücs et al. [64–66], is used to determine the unknown parameters. Solutions are given in Tables A4 and A5.

$$\min_{\substack{a_i, b_i, r_i \\ i=1}} \left[ \sum_{i=1}^{N} \left( \int_{r_{i-1}}^{r_i} (f(z) - (a_i z + b_i))^2 dz \right) \right]$$
(A4)
such that
$$z_{\min} \le r_1 \le \dots \le r_{N-1} \le z_{\max}$$

$$a_i r_i + b_i = a_{i+1} r_i + b_{i+1}$$

$$r_0 = z_{\min}$$

$$r_N = z_{\max}.$$

Consequently, total ageing in the epigraph formulation is given by Equations (A5) and (A6).

$$\Delta SOH(k) = \Delta SOH^{-}(k) + \Delta SOH^{+}_{PWA}(k) \qquad \forall k \in \mathbb{I}_{[1:N_{k}]}$$
(A5)

$$a_i P_{\text{AC}}^+(k) + b_i \le \Delta SOH_{\text{PWA}}^+(k) \qquad \qquad \forall i \in \mathbb{I}_{[1:N]} \quad \forall k \in \mathbb{I}_{[1:N_h]}. \tag{A6}$$

In this formulation,  $\Delta SOH_{PWA}^+(k) \in \mathbb{R}^{N_h}$  constitutes the upper bound for the PWA functions at each time step k, similar to the role of  $f_{PWA}$  in the epigraph formulation given by Equation (A3).

#### Appendix A.2. Battery Loss Modelling

S

In battery loss modelling, two loss elements are considered: Battery internal losses and power electronic losses [14]. In this study, three cases are analysed: All losses are neglected (Case 1); only battery internal losses and ageing are considered (Case 2); and battery internal losses, battery ageing, and power-electronic losses are considered (Case 3). Therefore, as illustrated in Figure A7, there exist different data-driven functions that map the input power  $P_{AC}^{+/-}$  to the dissipation losses  $L_{bat}^{+/-}$  (battery dissipation losses),  $L_{PE}^{+/-}$  (power electronic losses) and  $L_{tot}^{+/-}$  (total system losses). Loss mapping for Cases 2 and 3 is shown in Figure A7, and  $L_{bat}^{+/-} = 0$  for Case 1. Therefore, according to the three different cases we study, there are four different functions  $f_{bat}^+$ ,  $f_{bat}^-$ ,  $f_{tot}^+$ , and  $f_{tot}^-$  (given in Figure A8) to be approximated with the method given in the previous subsection.



Figure A7. System Diagram.

Loss [kW]

5

10

15

0



Figure A8. Loss values for different AC side power levels.

20

 $P_{AC}$  [kW]

25

30

35

40

45

Compared to the PWA approximation of the ageing function, which is an increasing convex function, there are two complications for the modelling of the loss functions:

- 1. Some of the price values may be negative, which causes the loss function to be concave at that time step; therefore, the epigraph formulation given in Equation (A3) cannot be used.
- 2. Although the battery loss functions  $f_{bat}^+$  and  $f_{bat}^-$  are convex, the combined loss functions  $f_{tot}^+$  and  $f_{tot}^-$  are non-convex. Hence, they can not be written as in Equation (A3).

Due to the above problems, the representation in Equation (A3) cannot be completely relied on for selecting the appropriate part of the PWA function. Therefore, we need to expand our formulation into the mixed-integer domain to include binary variables, which can be utilised to select the correct segment of the PWA function. However, despite having the state-of-the-art MILP solvers, excessive use of integer variables may lead to an insufficient memory problem and an increased computation time. Hence, we utilise a decomposition method that will reduce the number of redundant binary and real variables. The procedure followed is produced in Figures A9 and A10.

The procedure given in Figures A9 and A10 is basically a classifier that makes rough decisions to determine if a part of the system is convex, non-convex, or concave, which are collected, respectively, in the sets  $\mathcal{E}$ ,  $\mathcal{B}$ , and  $\mathcal{A}$ . The algorithm assumes that the PWA approximations are mostly convex, and that there are only several pieces that violate the convexity condition. The validity of this assumption can be verified by Figure A8. Only a small portion of  $L_{tot}^{+/-}$  causes non-convexity. Therefore, using binary variables to represent all segments is not necessary. For this reason, the function can be split into several regions that are convex themselves and fewer binary variables that can be utilised to select amongst these regions instead of all piece-wise functions. Inside each different region, the epigraph formulation can be leveraged to select the correct piece. On the other hand, as given in Equation (7), the cost incurred by the efficiency losses is embedded in the first term, and it can be expanded into Equation (A7) by using Equations (10)–(12):

$$c_{en}(k) \cdot P_{AC}(k)\Delta t = c_{en}(k) \cdot (P_{AC}^{+}(k) - P_{AC}^{-}(k))\Delta t$$
  
=  $c_{en}(k) \cdot ([P_{b}^{+}(k) + L^{+}(k)] - [P_{b}^{-}(k) - L^{-}(k)])\Delta t$   
=  $c_{en}(k) \cdot ([P_{b}^{+}(k) - P_{b}^{-}(k)] + [L^{+}(k) + L^{-}(k)])\Delta t$   
=  $c_{en}(k) \cdot \underbrace{[P_{b}^{+}(k) - P_{b}^{-}(k)]}_{independent decision variables} \cdot \Delta t + c_{en}(k) \cdot \underbrace{[L^{+}(k) + L^{-}(k)]}_{approximated variables} \cdot \Delta t.$  (A7)



Figure A9. Flowchart to determine approximation method for losses.



Figure A10. Flowchart for Algorithm 2.

For the epigraph formulation to hold, the approximated variables, shown in Equation (A7), must be minimised in the optimisation problem. However, if the multiplier  $c_{en}(k)$  at time k is negative, the loss values in the brackets are going to be maximised by the optimisation solver. In other words, having a negative price signal at time k causes the curves shown in Figure A8 to be reversed and become concave or mostly concave. Therefore, a full-scale MILP formulation with binary variables assigned for each piece in the PWA approximation must be used. In summary, when the price signal is negative, the loss function is multiplied by a negative number and converted into a mostly concave function. In this case, the algorithm executes the complete MILP modelling at that time step. If the price signal is not negative, but the PWA approximation violates the convexity requirement at some points (e.g., the slope is not monotonically increasing), then the algorithm collects all breakpoints and the time step in the set  $\mathcal{B}$ . Lastly, if the PWA approximation is completely convex, then the procedure given in Appendix A.1 (the previous section) is applied.

After the problem separation, the final equations to complete the system formulation are given. First, the *a*, *b*, and *r* coefficients and breakpoints must be defined in the system, according to the studied case (where the numerical values are presented in Table A4). Selected parameters for charge and discharge will be called  $a^+$ ,  $b^+$ , and  $r^+$ , and  $a^-$ ,  $b^-$ , and  $r^-$ , respectively. Then, by leveraging the algorithm given in Figure A9 for both charge and discharge approximations, the sets  $\mathcal{E}^+$ ,  $\mathcal{B}^+$ , and  $\mathcal{A}^+$ , and  $\mathcal{E}^-$ ,  $\mathcal{B}^-$ , and  $\mathcal{A}^-$  are obtained, respectively. Then, we define two group of variables for the full MILP and economic MILP models. Binary variables are shown by *S*, where continuous variables are given by *P*, with indices full and eco for full MILP and economic MILP, respectively. finally, the variables related to charging and discharging are indicated by a plus and minus sign as superscript, respectively. Therefore, the newly defined variables are given by  $S^+_{full}(k,j)$ ,  $S^+_{eco}(k,j) \in \mathbb{Z}_2$ ,  $P^+_{full}(k,j)$ ,  $P^-_{eco}(k,j) \in \mathbb{R}$ , in respective sets of appropriate dimensions. Thereafter, we define the following constraints. For all *k* that the epigraph formulation can be used, the constraints are defined by Equations (A8) and (A9).

$$L^{+}(k) \ge a_{j}^{+} P_{AC}^{+}(k) + b_{j}^{+} \qquad \qquad \forall j \in \mathbb{I}_{[1:N]} \quad \forall k \in \mathcal{E}^{+}$$
(A8)

$$L^{-}(k) \ge a_{j}^{-} P_{AC}^{-}(k) + b_{j}^{-} \qquad \qquad \forall j \in \mathbb{I}_{[1:N]} \quad \forall k \in \mathcal{E}^{-}.$$
(A9)

For all *k* that the full MILP formulation must be used, the constraints are defined by Equations (A10)–(A15).

$$L^{+}(k) = \sum_{j} \left( a_{j}^{+} P_{full}^{+}(k,j) + b_{j}^{+} S_{full}^{+}(k,j) \right) \qquad \forall j \in \mathbb{I}_{[1:N]} \quad \forall k \in \mathcal{A}^{+}$$
(A10)

$$L^{-}(k) = \sum_{j} \left( a_{j}^{-} P_{full}^{-}(k, j) + b_{j}^{-} S_{full}^{-}(k, j) \right) \qquad \forall j \in \mathbb{I}_{[1:N]} \quad \forall k \in \mathcal{A}^{-}$$
(A11)

$$S^{+}_{full}(k,j)r^{+}_{j-1} \le P^{+}_{full}(k,j) \le S^{+}_{full}(k,j)r^{+}_{j} \qquad \forall j \in \mathbb{I}_{[1:N]} \quad \forall k \in \mathcal{A}^{+}$$
(A12)

$$S_{full}^{-}(k,j)r_{j-1}^{-} \le P_{full}^{-}(k,j) \le S_{full}^{-}(k,j)r_{j}^{-} \qquad \forall j \in \mathbb{I}_{[1:N]} \quad \forall k \in \mathcal{A}^{-}$$
(A13)

$$\sum_{j} \left( S_{full}^{+}(k,j) \right) = 1 \qquad \qquad \forall j \in \mathbb{I}_{[1:N]} \quad \forall k \in \mathcal{A}^{+} \qquad (A14)$$

$$\sum_{j} \left( S_{full}^{-}(k,j) \right) = 1 \qquad \qquad \forall j \in \mathbb{I}_{[1:N]} \quad \forall k \in \mathcal{A}^{-}.$$
(A15)

For all k that the economic MILP formulation can be utilised, the constraints are defined by Equations (A16)–(A21).

$$L^{+}(k) \geq \sum_{v=p_{1}^{+}}^{p_{2}^{+}} \left( a_{v} P_{eco}^{+}(k,i) + b_{v} S_{eco}^{+}(k,i) \right)$$
  
$$\forall i \in \mathcal{B}_{I}^{+} \land \forall k \in \mathcal{B}_{K}^{+} \land (k, \mathcal{B}_{I}^{+}, \mathcal{B}_{J}^{+}) \in \mathcal{B}^{+} \land (k, p_{1}^{+}, p_{2}^{+}) \in \mathcal{B}_{J}^{+}$$
(A16)  
$$p_{2}^{-}$$

$$L^{-}(k) \geq \sum_{v=p_{1}^{-}}^{P_{2}} \left( a_{v} P_{eco}^{-}(k,i) + b_{v} S_{eco}^{-}(k,i) \right)$$
  
$$\forall i \in \mathcal{B}_{I}^{-} \land \forall k \in \mathcal{B}_{K}^{-} \land (k, \mathcal{B}_{I}^{-}, \mathcal{B}_{J}^{-}) \in \mathcal{B}^{-} \land (k, p_{1}^{-}, p_{2}^{-}) \in \mathcal{B}_{J}^{-}$$
(A17)

$$S_{eco}^{+}(k,i)r_{(p_{1}^{+}-1)}^{+} \leq P_{eco}^{+}(k,i) \leq S_{eco}^{+}(k,i)r_{p_{2}^{+}}^{+}$$

$$\forall i \in \mathcal{B}_{I}^{+} \land \forall k \in \mathcal{B}_{K}^{+} \land (k, \mathcal{B}_{I}^{+}, \mathcal{B}_{J}^{+}) \in \mathcal{B}^{+} \land (k, p_{1}^{+}, p_{2}^{+}) \in \mathcal{B}_{J}^{+}$$

$$S_{eco}^{-}(k, i)r_{(p_{1}^{-}-1)}^{-} \leq P_{eco}^{-}(k, i) \leq S_{eco}^{-}(k, i)r_{p_{2}^{-}}^{-}$$

$$(A18)$$

$$\forall i \in \mathcal{B}_I^- \land \forall k \in \mathcal{B}_K^- \land (k, \mathcal{B}_I^-, \mathcal{B}_J^-) \in \mathcal{B}^- \land (k, p_1^-, p_2^-) \in \mathcal{B}_J^-$$
(A19)

$$\sum_{i} \left( S^{+}_{eco}(k,i) \right) = 1 \quad \forall i \in \mathcal{B}^{+}_{I} \land \forall k \in \mathcal{B}^{+}_{K} \land (k, \mathcal{B}^{+}_{I}, \mathcal{B}^{+}_{J}) \in \mathcal{B}^{+}$$
(A20)

$$\sum_{i} \left( S_{eco}^{-}(k,i) \right) = 1 \quad \forall i \in \mathcal{B}_{I}^{-} \land \forall k \in \mathcal{B}_{K}^{-} \land (k,\mathcal{B}_{I}^{-},\mathcal{B}_{J}^{-}) \in \mathcal{B}^{-}.$$
(A21)

**Table A4.** PWA approximation parameters for power loss ( $r_0 = 0$  is omitted).

| Efficiency         |               |       |                    |         |             |                    |         |        |                    |         |       |
|--------------------|---------------|-------|--------------------|---------|-------------|--------------------|---------|--------|--------------------|---------|-------|
| Charging           |               |       |                    |         | Discharging |                    |         |        |                    |         |       |
|                    | Case 2 Case 3 |       |                    |         | Case 2      |                    |         | Case 3 |                    |         |       |
| а                  | b             | r     | а                  | b       | r           | а                  | b       | r      | а                  | b       | r     |
| $(\times 10^{-3})$ | [kW]          | [kW]  | $(\times 10^{-3})$ | [kW]    | [kW]        | $(\times 10^{-3})$ | [kW]    | [kW]   | $(\times 10^{-3})$ | [kW]    | [kW]  |
| 3.29               | 0             | 4.26  | 1000               | 0       | 0.29        | 4.71               | 0       | 4.68   | 467.65             | 0       | 0.44  |
| 14.99              | -0.0498       | 9.57  | 12.14              | 0.2655  | 5.33        | 19.31              | -0.0684 | 10.34  | 24.18              | 0.1955  | 7.74  |
| 26.26              | -0.1577       | 14.94 | 30.3               | 0.1688  | 10.97       | 33.98              | -0.2202 | 15.93  | 39.5               | 0.077   | 13.3  |
| 37.23              | -0.3216       | 20.39 | 46.17              | -0.0052 | 16.95       | 48.89              | -0.4576 | 21.46  | 58.97              | -0.1821 | 19.47 |
| 47.97              | -0.5406       | 25.97 | 61.36              | -0.2627 | 22.96       | 64.07              | -0.7835 | 26.94  | 78.7               | -0.5662 | 25.66 |
| 58.56              | -0.8155       | 31.68 | 76.69              | -0.6147 | 28.94       | 79.74              | -1.2054 | 32.41  | 101.55             | -1.1526 | 31.62 |
| 68.86              | -1.1419       | 37.39 | 91.77              | -1.0513 | 35.17       | 95.58              | -1.7188 | 37.84  | 118.59             | -1.6915 | 37.56 |
| 79.01              | -1.5213       | 43.2  | 104.71             | -1.5062 | 43.2        | 111.67             | -2.3276 | 43.2   | 135.27             | -2.3179 | 43.2  |

**Table A5.** PWA approximation parameters for cycle ageing ( $r_0 = 0$  is omitted).

| Cycle Ageing              |                        |       |  |  |  |
|---------------------------|------------------------|-------|--|--|--|
| a (×10 <sup>-8</sup> )    | b (×10 <sup>-6</sup> ) | r     |  |  |  |
| $[\Delta SOH \cdot h/kW]$ | $[\Delta SOH]$         | [kW]  |  |  |  |
| 1.44                      | 0                      | 9.24  |  |  |  |
| 2.97                      | -0.142                 | 17.11 |  |  |  |
| 5.62                      | -0.594                 | 22.8  |  |  |  |
| 9.58                      | -1.5                   | 27.25 |  |  |  |
| 15.1                      | -3.01                  | 30.94 |  |  |  |
| 22.5                      | -5.28                  | 34.06 |  |  |  |
| 31.8                      | -8.47                  | 36.76 |  |  |  |
| 43.4                      | -12.7                  | 39.13 |  |  |  |
| 57.3                      | -18.1                  | 41.26 |  |  |  |
| 73.7                      | -24.9                  | 43.2  |  |  |  |

## References

- Hesse, H.; Schimpe, M.; Kucevic, D.; Jossen, A. Lithium-Ion Battery Storage for the Grid—A Review of Stationary Battery Storage System Design Tailored for Applications in Modern Power Grids. *Energies* 2017, 10, 2107. [CrossRef]
- 2. Graves, F.; Jenkin, T.; Murphy, D. Opportunities for electricity storage in deregulating markets. *Electr. J.* **1999**, *12*, 46–56. [CrossRef]
- Kondoh, J.; Ishii, I.; Yamaguchi, H.; Murata, A.; Otani, K.; Sakuta, K.; Higuchi, N.; Sekine, S.; Kamimoto, M. Electrical energy storage systems for energy networks. *Energy Convers. Manag.* 2000, 41, 1863–1874. [CrossRef]
- 4. Schoenung, S.M.; Burns, C. Utility energy storage applications studies. *IEEE Trans. Energy Convers.* **1996**, *11*, 658–665. [CrossRef]
- 5. Hirth, L.; Ueckerdt, F.; Edenhofer, O. Integration costs revisited—An economic framework for wind and solar variability. *Renew. Energy* **2015**, *74*, 925–939. [CrossRef]
- 6. Dunn, B.; Kamath, H.; Tarascon, J.M. Electrical Energy Storage for the Grid: A Battery of Choices. *Science* **2011**, 334, 928–935. [CrossRef] [PubMed]
- 7. Greenwood, D.M.; Lim, K.Y.; Patsios, C.; Lyons, P.F.; Lim, Y.S.; Taylor, P.C. Frequency response services designed for energy storage. *Appl. Energy* **2017**, *203*, 115–127. [CrossRef]
- Mantar Gundogdu, B.; Nejad, S.; Gladwin, D.T.; Foster, M.P.; Stone, D.A. A Battery Energy Management Strategy for U.K. Enhanced Frequency Response and Triad Avoidance. *IEEE Trans. Ind. Electron.* 2018, 65, 9509–9517. [CrossRef]
- 9. Moreno, R.; Moreira, R.; Strbac, G. A MILP model for optimising multi-service portfolios of distributed energy storage. *Appl. Energy* **2015**, *137*, 554–566. [CrossRef]
- 10. Fong, G.; Moreira, R.; Strbac. G. Economic analysis of energy storage business models. In Proceedings of the 2017 IEEE Manchester PowerTech, Manchester, UK, 18–22 June 2017.
- 11. F. Sorourifar.; V. M. Zavala.; A. W. Dowling. Integrated Multiscale Design, Market Participation, and Replacement Strategies for Battery Energy Storage Systems. *IEEE Trans. Sustain. Energy* **2018**. [CrossRef]
- 12. Stephan, A.; Battke, B.; Beuse, M.D.; Clausdeinken, J.H.; Schmidt, T.S. Limiting the public cost of stationary battery deployment by combining applications. *Nat. Energy* **2016**, *1*, 16079. [CrossRef]
- 13. Shi, Y.; Xu, B.; Wang, D.; Zhang, B. Using Battery Storage for Peak Shaving and Frequency Regulation: Joint Optimization for Superlinear Gains. *IEEE Trans. Power Syst.* **2018**, *33*, 2882–2894. [CrossRef]
- Schimpe, M.; Truong, C.N.; Naumann, M.; Jossen, A.; Hesse, H.C.; Reniers, J.M.; Howey, D.A. Marginal Costs of Battery System Operation in Energy Arbitrage Based on Energy Losses and Cell Degradation. In Proceedings of the 2018 IEEE International Conference on Environment and Electrical Engineering and 2018 IEEE Industrial and Commercial Power Systems Europe (EEEIC/I&CPS Europe), Palermo, Italy, 12–15 June 2018; IEEE: Piscataway, NJ, USA, 2018; pp. 1–5.
- 15. Bradbury, K.; Pratson, L.; Patiño-Echeverri, D. Economic viability of energy storage systems based on price arbitrage potential in real-time U.S. electricity markets. *Appl. Energy* **2014**, *114*, 512–519. [CrossRef]
- Koller, M.; Borsche, T.; Ulbig, A.; Andersson, G. Defining a degradation cost function for optimal control of a battery energy storage system. In Proceedings of the 2013 IEEE Grenoble Conference, Grenoble, France, 16–20 June 2013; IEEE: Piscataway, NJ, USA, 2013; pp. 1–6.
- Stroe, D.I.; Swierczynski, M.; Stroe, A.I.; Laerke, R.; Kjaer, P.C.; Teodorescu, R. Degradation Behavior of Lithium-Ion Batteries Based on Lifetime Models and Field Measured Frequency Regulation Mission Profile. *IEEE Trans. Ind. Appl.* 2016, *52*, 5009–5018. [CrossRef]
- Swierczynski, M.; Stroe, D.I.; Stan, A.I.; Teodorescu, R.; Sauer, D.U. Selection and Performance-Degradation Modeling of LiMO<sub>2</sub>/Li<sub>4</sub>Ti<sub>5</sub>O<sub>12</sub> and LiFePO<sub>4</sub>/C Battery Cells as Suitable Energy Storage Systems for Grid Integration With Wind Power Plants: An Example for the Primary Frequency Regulation Service. *IEEE Trans. Sustain. Energy* 2014, *5*, 90–101. [CrossRef]
- 19. Sun, S.; Dong, M.; Liang, B. Real-time power balancing in electric grids with distributed storage. *IEEE J. Sel. Top. Signal Process.* **2014**, *8*, 1167–1181. [CrossRef]
- 20. Sarker, M.R.; Murbach, M.D.; Schwartz, D.T.; Ortega-Vazquez, M.A. Optimal operation of a battery energy storage system: Trade-off between grid economics and storage health. *Electr. Power Syst. Res.* 2017, 152, 342–349. [CrossRef]

- 21. Xu, B.; Zhao, J.; Zheng, T.; Litvinov, E.; Kirschen, D.S. Factoring the Cycle Aging Cost of Batteries Participating in Electricity Markets. *IEEE Trans. Power Syst.* **2018**, *33*, 2248–2259. [CrossRef]
- 22. Goebel, C.; Hesse, H.; Schimpe, M.; Jossen, A.; Jacobsen, H.A. Model-Based Dispatch Strategies for Lithium-Ion Battery Energy Storage Applied to Pay-as-Bid Markets for Secondary Reserve. *IEEE Trans. Power Syst.* 2017, *32*, 2724–2734. [CrossRef]
- 23. Perez, A.; Moreno, R.; Moreira, R.; Orchard, M.; Strbac, G. Effect of battery degradation on multi-service portfolios of energy storage. *IEEE Trans. Sustain. Energy* **2016**, *7*, 1718–1729. [CrossRef]
- 24. Patsios, C.; Wu, B.; Chatzinikolaou, E.; Rogers, D.J.; Wade, N.; Brandon, N.P.; Taylor, P. An integrated approach for the analysis and control of grid connected energy storage systems. *J. Energy Storage* **2016**, *5*, 48–61. [CrossRef]
- Lee, S.B.; Pathak, C.; Ramadesigan, V.; Gao, W.; Subramanian, V.R. Direct, efficient, and real-time simulation of physics-based battery models for stand-alone pv-battery microgrids. *J. Electrochem. Soc.* 2017, *164*, E3026–E3034. [CrossRef]
- Weißhar, B.; Bessler, W.G. Model-based degradation assessment of lithium-ion batteries in a smart microgrid. In Proceedings of the 2015 International Conference on Smart Grid and Clean Energy Technologies (ICSGCE), Offenburg, Germany, 20–23 October 2015.
- 27. Reniers, J.M.; Mulder, G.; Ober-Blöbaum, S.; Howey, D.A. Improving optimal control of grid-connected lithium-ion batteries through more accurate battery and degradation modelling. *J. Power Sources* **2018**, 379, 91–102. [CrossRef]
- 28. Schimpe, M.; Naumann, M.; Truong, N.; Hesse, H.C.; Santhanagopalan, S.; Saxon, A.; Jossen, A. Energy efficiency evaluation of a stationary lithium-ion battery container storage system via electro-thermal modeling and detailed component analysis. *Appl. Energy* **2018**, *210*, 211–229. [CrossRef]
- 29. Schimpe, M.; Piesch, C.; Hesse, H.; Paß, J.; Ritter, S.; Jossen, A. Power Flow Distribution Strategy for Improved Power Electronics Energy Efficiency in Battery Storage Systems: Development and Implementation in a Utility-Scale System. *Energies* **2018**, *11*, 533. [CrossRef]
- Schimpe, M.; Becker, N.; Lahlou, T.; Hesse, H.C.; Herzog, H.G.; Jossen, A. Energy efficiency evaluation of grid connection scenarios for stationary battery energy storage systems. *Energy Procedia* 2018, 155, 77–101. [CrossRef]
- 31. Gurobi Optimization, Inc. *Gurobi Optimizer Reference Manual Version 8.1;* Gurobi Optimization: Houston, TX, USA, 2018.
- 32. Meindl, B.; Templ, M. *Analysis of Commercial and Free and Open Source Solvers for Linear Optimization Problems*; Tech. Rep. CS-2012-1; Dept. Inst. Statist., Vienna University of Technology: Vienna, Austria, 2012; Volume 20.
- Bixby, R.E. A Brief History of Linear and Mixed-Integer Programming Computation; Documenta Math., Extra Volume: Optimization Stories; 2012; pp. 107–121. Available online: https://www.math.uni-bielefeld.de/ documenta/vol-ismp/25\_bixby-robert.pdf (accessed on 13 March 2019)
- Klotz, E.; Newman, A.M. Practical guidelines for solving difficult mixed integer linear programs. *Surv. Oper. Res. Manag. Sci.* 2013, 18, 18–32. [CrossRef]
- 35. Cheng, B.; Powell, W.B. Co-optimizing battery storage for the frequency regulation and energy arbitrage using multi-scale dynamic programming. *IEEE Trans. Smart Grid* **2018**, *9*, 1997–2005. [CrossRef]
- 36. Krishnamurthy, D.; Uckun, C.; Zhou, Z.; Thimmapuram, P.R.; Botterud, A. Energy storage arbitrage under day-ahead and real-time price uncertainty. *IEEE Trans. Power Syst.* **2018**, *33*, 84–93. [CrossRef]
- 37. Nikolakakis, T.; Fthenakis, V. Compressed Air Energy Storage Models for Energy Arbitrage and Ancillary Services: Comparison Using Mixed Integer Programming Optimization with Market Data from the Irish Power System. *Energy Technol.* **2018**, *6*, 1290–1301. [CrossRef]
- 38. Pawel, I. The Cost of Storage—How to Calculate the Levelized Cost of Stored Energy (LCOE) and Applications to Renewable Energy Generation. *Energy Procedia* **2014**, *46*, 68–77. [CrossRef]
- 39. Yasuda, M. Sony energy storage system using olivine type battery. In Proceedings of the International Conference on Olivines for Rechargeable Batteries, Montreal, QC, Canada, 25–28 May 2014.
- 40. Schimpe, M.; von Kuepach, M.E.; Naumann, M.; Hesse, H.C.; Smith, K.; Jossen, A. Comprehensive Modeling of Temperature-Dependent Degradation Mechanisms in Lithium Iron Phosphate Batteries. *J. Electrochem. Soc.* **2018**, *165*, A181–A193. [CrossRef]
- 41. Naumann, M.; Schimpe, M.; Keil, P.; Hesse, H.C.; Jossen, A. Analysis and modeling of calendar aging of a commercial LiFePO4/graphite cell. *J. Energy Storage* **2018**, *17*, 153–169. [CrossRef]

- 42. Schmalstieg, J.; Käbitz, S.; Ecker, M.; Sauer, D.U. A holistic aging model for Li (NiMnCo) O2 based 18650 lithium-ion batteries. *J. Power Sources* **2014**, 257, 325–334. [CrossRef]
- 43. mg Solar. Product Pricing Information for Sony Fortelion Module. 2019. Available online: www.mg-solar-shop.com/pv-battery-offgrid-systems/sony-fortelion-life-po4-energy-storage-module-1-2-kw.html (accessed on 2 February 2019).
- 44. Nykvist, B.; Nilsson, M. Rapidly falling costs of battery packs for electric vehicles. *Nat. Clim. Chang.* **2015**, *5*, 329–332. [CrossRef]
- 45. EPEX Spot. EPEX Intraday and Day Ahead Market Data. Available online: https://www.epexspot.com/ de/marktdaten/ (accessed on 25 December 2018).
- 46. Zeh, A.; Müller, M.; Naumann, M.; Hesse, H.; Jossen, A.; Witzmann, R. Fundamentals of Using Battery Energy Storage Systems to Provide Primary Control Reserves in Germany. *Batteries* **2016**, *2*, 29. [CrossRef]
- 47. Transmission Grid Operators. 2019. Available online: https://www.regelleistung.net/ext/tender/ (accessed on 2 February 2019).
- Fleer, J.; Zurmühlen, S.; Meyer, J.; Badeda, J.; Stenzel, P.; Hake, J.F.; Uwe Sauer, D. Price development and bidding strategies for battery energy storage systems on the primary control reserve market. *Energy Procedia* 2017, 135, 143–157. [CrossRef]
- 49. CPLEX, IBM ILOG V12. 1: User's Manual for CPLEX. *Int. Bus. Mach. Corp.* **2009**, *46*, 157. Available online: http://www.cplex.com (accessed on 2 December 2018).
- 50. Houska, B.; Ferreau, H.J.; Diehl, M. ACADO toolkit—An open-source framework for automatic control and dynamic optimization. *Optim. Control Appl. Methods* **2011**, *32*, 298–312. [CrossRef]
- 51. Kocer, B.B.; Tjahjowidodo, T.; Seet, G.G.L. Centralized predictive ceiling interaction control of quadrotor VTOL UAV. *Aerosp. Sci. Technol.* **2018**, *76*, 455–465. [CrossRef]
- 52. Kumtepeli, V.; Zhao, Y.; Naumann, M.; Tripathi, A.; Wang, Y.; Jossen, A.; Hesse, H.C. Design and Analysis of an Aging-Aware Energy Management System for Islanded Grids Using Mixed-Integer Quadratic Programming. *Int. J. Energy Res.* 2019, 1–18, unpublished.
- 53. Martins, R.; Hesse, H.; Jungbauer, J.; Vorbuchner, T.; Musilek, P. Optimal component sizing for peak shaving in battery energy storage system for industrial applications. *Energies* **2018**, *11*, 2048. [CrossRef]
- 54. Löfberg, J. YALMIP : A Toolbox for Modeling and Optimization in MATLAB. In Proceedings of the CACSD Conference, Taipei, Taiwan, 2–4 September 2004.
- 55. Gurobi Optimization Inc. *Gurobi Optimizer Version 8.1;* Gurobi Optimization, Inc.: Houston, TX, USA, 22 October 2018 (Software Program). Available online: http://www.gurobi.com (accessed on 7 December 2018).
- 56. Schuster, S.F.; Bach, T.; Fleder, E.; Müller, J.; Brand, M.; Sextl, G.; Jossen, A. Nonlinear aging characteristics of lithium-ion cells under different operational conditions. *J. Energy Storage* **2015**, *1*, 44–53. [CrossRef]
- Zhao, X.; Yin, Y.; Hu, Y.; Choe, S.Y. Electrochemical-thermal modeling of lithium plating/stripping of Li(Ni0.6Mn0.2Co0.2)O2/Carbon lithium-ion batteries at subzero ambient temperatures. *J. Power Sources* 2019, 418, 61–73. [CrossRef]
- 58. Kocer, B.B.; Tjahjowidodo, T.; Seet, G.G.L. Model predictive uav-tool interaction control enhanced by external forces. *Mechatronics* **2019**, *58*, 47–57. [CrossRef]
- 59. Kumtepeli, V.; Wang, Y.; Tripathi, A. Multi-area model predictive load frequency control: A decentralized approach. In Proceedings of the Asian Conference on Energy, Power and Transportation Electrification (ACEPT 2016), Singapore, 25–27 October 2016; IEEE: Piscataway, NJ, USA, 2016; pp. 1–5.
- 60. Müller, F.L.; Woerner, S.; Lygeros, J. Ramp-Rate-Constrained Bidding of Energy and Frequency Reserves in Real Market Settings. *arXiv* **2018**, arXiv:1804.03892.
- 61. Hempel, A.B.; Goulart, P.J.; Lygeros, J. Every continuous piecewise affine function can be obtained by solving a parametric linear program. In Proceedings of the 2013 European Control Conference (ECC), Zurich, Switzerland, 17–19 July 2013; pp. 2657–2662.
- 62. Hempel, A.B.; Goulart, P.J.; Lygeros, J. Inverse parametric optimization with an application to hybrid system control. *IEEE Trans. Autom. Control* **2015**, *60*, 1064–1069. [CrossRef]
- 63. Boyd, S.; Vandenberghe, L. Convex Optimization; Cambridge University Press: Cambridge, MA, USA, 2004.
- 64. Kvasnica, M.; Szucs, A.; Fikar, M. Automatic derivation of optimal piecewise affine approximations of nonlinear systems. *IFAC Proc. Vol.* **2011**, *44*, 8675–8680. [CrossRef]

- 65. Števek, J.; Szucs, A.; Kvasnica, M.; Kozák, Š.; Fikar, M. Smart technique for identifying hybrid systems. In Proceedings of the 2012 IEEE 10th International Symposium on Applied Machine Intelligence and Informatics (SAMI), Herl'any, Slovakia, 26–28 January 2012; pp. 383–388.
- 66. Szucs, A.; Kvasnica, M.; Fikar, M. Optimal Piecewise Affine Approximations of Nonlinear Functions Obtained from Measurements. *IFAC Proc. Vol.* **2012**, *45*, 160–165. [CrossRef]



© 2019 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).